Author's response to reviews

Title: Knowledge, Attitudes and Practices of grade three primary schoolchildren in relation to schistosomiasis, soil transmitted helminthiasis and malaria in Zimbabwe

Authors:

Nicholas Midzi (midzinicholas@yahoo.com)
Sekesai Mtapuri-Zinyowera (sekesaiz@yahoo.com)
Munyaradzi P Mapingure (mmapingure@yahoo.com)
Noah H Paul (npaul@science.uz.ac.zw)
Davison Sangweme (dsangweme@yahoo.com)
Gibson Hlerema (ghlerema@gmail.com)
Masciline J Mutsaka (mjmutsaka@yahoo.com)
Farisai Tongogara (ftongogara@gmail.com)
Godfrey Makware (gmakware@yahoo.com)
Vivian Chadukura (vchadukura@yahoo.com)
Nirbhay Kumar (nkumar@jhsph.edu)
Francisca Mutapi (fmutapi@ed.ac.uk)
Takafira Mduluza (tmduluza@bhp.org.bw)

Version: 3 Date: 25 January 2011

Author's response to reviews:

The study has been registered with Clinical Trials.gov with the following details.
ClinicalTrials.gov Identifier: NCT01283165

I hope that the manuscript would be considered.